Author's response to reviews

Title: An exfoliation and enrichment strategy results in improved transcriptional profiles when compared to matched formalin fixed samples.

Authors:

    Wilfrido D Mojica (mojica@buffalo.edu)
    Leighton Stein (Leighton.Stein@RoswellPark.org)
    Lesleyann Hawthorn (Lesleyann.Hawthorn@RoswellPark.org)

Version: 4 Date: 31 July 2007

Author's response to reviews: see over
RE: BMC Clinical Pathology manuscript MS 1497657999135894 editorial revision

Dear Editors,

The following revisions have been incorporated into the submitted manuscript:
1) full stop at end of title
2) deleted single dot on page 3
3) “Figures” changed to “Figure legends”
4) stop removed from table title
5) highlighting / tracking changes removed
6) first letter of cited tables and figures capitalized
7) As requested by Scott Jewell, “Sednak” changed to “Sedmak”
8) As requested by Scott Jewel, redundant reference 17 deleted, and all changes updated in both text and reference page.

Thank you for your consideration and help.

Wilfrido D. Mojica, M.D.